Author's response to reviews

Title: "That never would have occurred to me": a qualitative study of medical students' views of a cultural competence curriculum

Authors:

Johanna Shapiro (jfshapir@uci.edu)
Desiree Lie (dalie@uci.edu)
David Gutierrez (dguiter@uci.edu)
Gabriella Zhuang (gzhuang@uci.edu)

Version: 5 Date: 24 May 2006

Author's response to reviews:

We have made the following editorial changes:
1) Capitalization of the letter 'M' in the title word "medical" is now lower-case.
2) The word count in the Abstract has been removed.
3) All authors for reference #7 are now included.